Author's response to reviews

Title: Effects of vildagliptin as compared to glibenclamide on glucose variability after a submaximal exercise test in patients with type 2 diabetes - DIABEX VILDA: study protocol for a randomized controlled trial

Authors:

Aline Aline Fofonka (alinefofonka@hotmail.com)
Jorge Pinto Ribeiro (alinefofonka@hotmail.com)
Karina Rabelo Casali (krabello@terra.com.br)
Beatriz D'Agord Schaan (beatrixschaan@gmail.com)

Version: 2
Date: 21 May 2014

Author's response to reviews: see over
Dr. Doug Altman
Editor-in-Chief of Trials

Attached to submission the revised manuscript entitled: “Effects of vildagliptin as compared to glibenclamide on glucose variability after a submaximal exercise test in patients with type 2 diabetes - DIABEX VILDA: study protocol for a randomized controlled trial”, which we ask to consider for publication in Trials.

The couple of matters that require are inserted on manuscript and described below:

- The information about informed consent form was inserted in the Ethical and data protection issues. “All patients will have to sign the informed consent form prior to participating in the study”.

- The key words were added after the abstract. “Key words: Diabetes mellitus; Hypoglycemic Agents; Exercise”.

Sincerely,
Beatriz D. Schaan MD, ScD
Endocrine Division, Hospital de Clínicas de Porto Alegre and School of Medicine, Universidade Federal do Rio Grande do Sul, Porto Alegre, RS, Brazil Rua Ramiro Barcelos, 2.350, prédio 12, 4º andar ;
Phone/Fax: +55 51 3359-8127
90035-003 Porto Alegre RS Brasil
beatrizschaan@gmail.com